Author's response to reviews

Title: Setting directions for capacity building in primary health care: a survey of a research network

Authors:

Karin Ried (karin.ried@flinders.edu.au)
Elizabeth A Farmer (liz.farmer@flinders.edu.au)
Kathryn M Weston (kath.weston@bigpond.com)

Version: 5 Date: 23 January 2006

Author's response to reviews:

Table 2 has been changed to portrait format